Author’s response to reviews

Title: REDUCING KIDNEY MOTION: OPTIMIZING ANESTHESIA AND COMBINING RESPIRATORY SUPPORT FOR RETROGRADE INTRARENAL SURGERY: A PILOT STUDY

Authors:

Nariman Gadzhiev (nariman.gadjiev@gmail.com)
Ullubiy Oibolatov (u.oibolatoff@yandex.ru)
Leonid Kolotilov (leon956@yandex.ru)
Sergei Parvanyan (parvanian@yandex.ru)
Gagik Akopyan (docgagik@mail.ru)
Sergei Petrov (petrov-uro@yandex.ru)
Courtney Cottone (ccottone@uci.edu)
John Sung (jsung7@uci.edu)
Zhamshid Okhunov (zokhunov@uci.edu)

Version: 2 Date: 28 Jun 2019

Author’s response to reviews:

Editor Comments:

1. Trial registration

Your study falls within the International Committee of Medical Journal Editors (ICMJE)’ definition of a clinical trial: any research study that prospectively assigns human subjects to one or more health related interventions to evaluate the effects on health or biological outcomes. As such, Biomed Central requires that a Trial Registration Number is provided in order for the manuscript to be published.

Although a TRN is usually required prior to the start of the peer-review process, the BMC-series journals does accept retrospectively registered trials. If you have not registered the trial, we therefore request that you do so as soon as possible so that your study can be accepted for publication. All trials must be registered with an ICMJE or WHO ICTRP approved registry, as listed in the ICMJE guide: http://www.icmje.org/faq.pdf, and on the WHO ICTRP website: http://www.who.int/ictrp/en/.
Once you know your trial registration number, please submit a revised version of your manuscript with the number and date of registration included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section.

AUTHOR RESPONSE: trial registration has been obtained and the number is shown at the end of the abstract.

2. Ethics approval
   a) full name of ethics committee

   Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate in the "Ethics Approval and Consent to Participate" subsection of the Declarations.

   AUTHOR RESPONSE: The full name of the ethics committee is Local Ethics committee as indicated in the manuscript. We have included the institute to which it belongs to.

   b) form of consent

   In the ethics approval and consent to participate statement, please confirm whether the consent you obtained from study participants was written or verbal. If verbal consent was obtained, please clearly state the reason or if this was approved by the ethics committee.

   AUTHOR RESPONSE: Written consent was obtained from all patients for study participation as well as for publication. This is stated in the methods section of the manuscript as well as in the declaration section.

3. Competing interests - Section Editor submission

   As a member of the editorial board (Section Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations and confirm that you had no role in the handling of this manuscript.

   AUTHOR RESPONSE: we have included a statement in the Declaration section of the manuscript that there is no competing interest from Dr. Okhunov in handling this manuscript.
4. Authors' contributions

The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

e.g.

ZO - study conception & design and manuscript writing
XX - data collection and analysis
XY - study conception and design

AUTHOR RESPONSE: Author contributions have been shown as requested.

5. Move Declarations

Please move the declarations section to after the abbreviations section and before the references.

AUTHOR RESPONSE: done.

6. Abstract formatting

Please format your abstract with the following headings: Background, Methods, Results, Conclusions.

AUTHOR RESPONSE: done

7. Consent for publication

Please clarify in the consent for publication statement whether the subject shown in figure 2 gave written consent for publication of this photograph.

AUTHOR RESPONSE: written consent was obtained as indicated in the methods section of the manuscript.

8. Figures
a) Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

AUTHOR RESPONSE: done

b) Figures should be attached to the manuscript as separate files, and each figure of a manuscript should be submitted as a single file.

AUTHOR RESPONSE: done.